Author’s response to reviews

Title: A case of severe hyponatremia: as a presenting feature of pituitary macroadenoma

Authors:

Kameera Bopeththa (Kameerabvkm@gmail.com)
Mohomed Niyaz (mniyas02@gmail.com)
Chathuranga Medagedara (senalmed@gmail.com)

Version: 4 Date: 24 Jan 2019

Author’s response to reviews:

Thank you for your comments. I have underlined the newly added sentences to the manuscript including past history, social history, family history, occupational history and details regarding alcohol and smoking. In addition, the manuscript was broken down into paragraphs.

The treatments received prior to the diagnosis was questioned. It is mentioned in the manuscript that, before the diagnosis of pituitary macroadenoma, he was simply on symptomatic treatment including 3% saline.

thank you